Author's response to reviews

Title: Corticosteroid co-treatment induces resistance to chemotherapy in surgical resections, xenografts, and established cell lines of pancreatic cancer

Authors:

Chengwen Zhang (C.ZHANG@DKFZ.DE)
Armin Kolb (ARMIN.KOLB@MED.UNI-HEIDELBERG.DE)
Peter Buchler (PETER.BUECHLER@MED.UNI-HEIDELBERG.DE)
Andrew C.B. Cato (ANDREW.CATO@ITG.FZK.DE)
Juergen Mattern (J.MATTERN@DKFZ.DE)
Werner Rittgen (W.RITTGEN@DKFZ.DE)
Lutz Edler (L.EDLER@DKFZ.DE)
Klaus-michael Debatin (KLAUS-MICHAEL.DEBATIN@MEDIZIN.UNI-ULM.DE)
Markus W Buchler (MARKUS.BUECHLER@MED.UNI-HEIDELBERG.DE)
Helmut Friess (HELMUT.FRIESS@MED.UNI-HEIDELBERG.DE)
Ingrid Herr (I.HERR@DKFZ.DE)

Version: 5 Date: 24 February 2006

Author’s response to reviews:

Dear Editors,

thank you for your mail from yesterday indicating acceptance of a revised version of our manuscript. We did now the appropriate changes in formatting as suggested in your letter.

Sincerely,

Ingrid Herr